Author’s response to reviews

Title: Adherence to, and effectiveness of an individually tailored home-based exercise program for frail older adults, driven by mobility monitoring: design of a prospective cohort study

Authors:

Hilde AE Geraedts (H.A.E.Geraedts@med.umcg.nl)
Wiebren Zijlstra (zijlstra@dshs-koeln.de)
Wei Zhang (wei.zhang01@philips.com)
Sjoerd K Bulstra (s.k.bulstra@umcg.nl)
Martin Stevens (m.stevens@umcg.nl)

Version: 2 Date: 22 May 2014

Author’s response to reviews:

Hilde Geraedts
Center for Human Movement Sciences
University of Groningen, University Medical Center Groningen
P.O. Box 196
9700 AD Groningen, The Netherlands
Tel. +31 (0) 50 363-8846
H.A.E.Geraedts@med.umcg.nl

To: The Editor-in-Chief, BMC Public Health
Concerns: Submission of manuscript

Groningen, 21 May 2014

Dear Sir/Madam,

Enclosed you will find our manuscript “Adherence to, and effectiveness of an individually tailored home-based exercise program for frail older adults, driven by mobility monitoring: design of a prospective cohort study”. This paper addresses the design and rationale of a study supported by The Netherlands Organisation for Health Research and Development (ZonMw), (funder approval number 40-00812-98-09014). We would be honored if you would consider this manuscript for your journal.

The manuscript has undergone several revisions at your request (21/05/2014):
1. The statement “Also, written informed consent will be completed at baseline.” is added to the Methods section (page 9, line 10-11).
2. The registration date has been added to the trial registration number at the end of the abstract (page 3, line 2).

3. Funding and ethics proofs as additional files have been removed.

I hereby state that all authors have read and approved submission of the manuscript, and that the material in the manuscript has not been published and is not being considered for publication elsewhere in whole or in part in any language.

Awaiting your reaction, on behalf of all co-authors,

I remain

Yours sincerely,

Hilde Geraedts, MSc.